# Virtual Mentor

American Medical Association Journal of Ethics March 2007, Volume 9, Number 3: 166-169.

#### Clinical case

# Diagnostic testing for diseases in the news

Commentary by Vidya Sharma, MBBS, MPH, and M. Denise Dowd, MD, MPH

Dr. Carpenter entered the exam room to see her next patient, Andrew, and his mother. Andrew was an 8-year-old-boy who was athletic and active. His mother was a bit of a worrier and frequently brought him to Dr. Carpenter for (generally) minor reasons.

During this visit, Andrew's mother pointed out a small rash on his right calf that had appeared the day before. Mrs. Wood said she was extremely concerned about the rash because of news reports of a flesh-eating bacteria spreading throughout the community. She believed the report mentioned that someone in Andrew's soccer league had come down with the infection.

Dr. Carpenter began her examination by asking Andrew about the rash. He said it was a little itchy, but not painful. He said he had not felt sick, and his mother reported that Andrew had not had a fever. During her examination, Dr. Carpenter noted no swelling, bleeding, pus or visibly dead tissue. She then explained to Andrew's mother that he had no symptoms consistent with the flesh-eating bacteria. But Mrs. Wood did not seem convinced, and she repeatedly emphasized that her son might have been exposed to the infection during soccer. She wanted more elaborate tests done, not satisfied with watching and waiting.

Realizing that this mother's anxiety could not be allayed, Dr. Carpenter debated about how to proceed. A proper diagnostic test for necrotizing fasciitis would require a biopsy, gram stain and culture. This procedure would require a lot of time and resources and seemed unnecessary in the absence of any typical symptoms. Dr. Carpenter also considered doing a rapid group A streptococcal test, which involves a simple throat swab that when placed on a reaction strip would give a positive or negative result. Although group A streptococcus causes necrotizing fasciitis, Dr. Carpenter was aware that there really were no recommendations for performing the rapid test for the suspected flesh-eating bacteria infection in Andrew's case.

Dr. Carpenter considered going through the full battery of tests to put Mrs. Wood's mind at rest.

#### **Commentary**

Dr. Carpenter is correct: there are no recommendations for doing a rapid throat swab test for necrotizing fasciitis (NF). Diagnostic testing for NF must be guided by the

history and physical examination; there is no reliable screening test to rule out the disease. Other diagnostic tests may include invasive and noninvasive diagnostics such as ultrasonography, CT scan, MRI, tissue oxygen saturation monitoring or biopsy [1-5], none of which is indicated in this case.

# Honoring parental requests and preferences for testing

Despite there being no medical indication of disease on physical examination, should Dr. Carpenter order further tests to appease her patient's mother? Patient involvement in medical decision making has dramatically changed in the past three decades, and respect for patient autonomy and patient preferences is now emphasized. A model of "enhanced" autonomy that respects the physician's professional opinion while encouraging patient involvement (but not patient control) is beginning to be favored [6]. Such an approach does not compel physicians to abdicate their professional responsibilities or clinical judgment [7]. Rather, physicians have an obligation to guide patients based on scientific evidence, clinical knowledge and experience. While patient preferences must be respected, physicians should balance them with the medical indications, costs, risks and benefits, and then act in a way that is in accord with their patients' interests and their own professional obligations.

Dr. Carpenter must first consider the potential for benefit versus possible harm that additional tests can cause. Getting a biopsy is invasive; it can cause pain, bleeding and scarring, and there is a low to nonexistent likelihood of benefit; other diagnostic tests can expose a patient to radiation and discomfort and can be very expensive without providing a definitive diagnosis. To be consistent with the principle of beneficence, Dr. Carpenter must inform Andrew's mother that further testing is not necessary and that she is not comfortable ordering any tests.

## Impact of the media on health services utilization

Popular media may influence physicians and patients in many ways [8]. In our case, a mother has heard about a condition on television, fears her son has that condition and seeks care for him. The impact of the media on the use of health services has been measured, and both increases [9] and decreases [10] in demand for services have been documented. Some authors have concluded that media coverage induces "an epidemic of anxiety," [11] resulting in decisions to alter the standard of care and producing a surge in diagnostic testing [9]. The accuracy of health-related stories in the media has been questioned by investigators who have reported widespread incomplete or inaccurate information [12, 13].

### **Ethics of overusing health care resources**

The overuse of health care resources is a rampant and well-recognized problem. One reason for this problem is physicians' concern about satisfying patient expectations. Other physician-related factors include fear of lawsuits, uncertainty and "standard of care" practices not based on evidence. While increased health care costs are an obvious negative side effect of unnecessary testing, a more direct implication—one of even greater concern—is the untoward physical and emotional consequences of

these tests on the patient. From a population perspective, overuse of resources (of which unnecessary testing is one example) can be considered less than ethical. The American College of Physicians Charter on Medical Professionalism explicitly states that

appropriate allocation of resources requires scrupulous avoidance of superfluous tests and procedures. The provision of unnecessary services not only exposes patients to avoidable harm and expense but also diminishes the resources available for others [14].

In short, performing tests that are not clinically indicated is unprofessional behavior.

#### Conclusion

The decision to pursue diagnostic testing should be based on evidence and assessment of the costs and benefits of the procedure. Obtaining a diagnostic test solely to allay patient anxiety with a negligible likelihood of medical benefit is not an ethically defensible practice from either individual or societal perspectives. The individual patient may suffer physical and emotional harms that outweigh any real or perceived benefit, and limited community resources will be inappropriately spent. The popular media is clearly a strong influence and source of knowledge for our patients. Harnessing the power of media through collaborative relationships with television, radio and print media outlets allows physicians to educate the public about health matters and lets the media serve as a primary outlet for information. As the source of expertise, however, the medical community has a responsibility to provide complete and accurate information in a timely manner.

# References

- 1. Yen ZS, Wang HP, Ma HM, Chen SC, Chen WJ. Ultrasonographic screening of clinically-suspected necrotizing fasciitis. *Acad Emerg Med.* 2002;9:1448-1451
- 2. Wall DB, Klein SR, Black S, de Virgilio C. A simple model to help distinguish necrotizing fasciitis from nonnecrotizing soft tissue infection. *J Am Coll Surg.* 2000;191:227-231.
- 3. Schmid MR, Kossmann T, Duewell S. Differentiation of necrotizing fasciitis and cellulitis using MR imaging. *AJR Am J Roentgenol*. 1998;170:615-620.
- 4. Wong CH, Khin LW, Heng KS, Tan KC, Low CO. The LRINEC (Laboratory Risk Indicator for Necrotizing Fasciitis) score: a tool for distinguishing necrotizing fasciitis from other soft tissue infections. *Crit Care Med*. 2004;32:1535-1541.
- 5. Wang TL, Hung CR. Role of tissue oxygen saturation monitoring in diagnosing necrotizing fasciitis of the lower limbs. *Ann Emerg Med*. 2004;44:222-228.
- 6. Hunink MGM, Glasziou PP, Siegel JE, et al. *Decision Making in Health and Medicine: Integrating Evidence and Values*. Cambridge, UK: Cambridge University Press; 2001.

- 7. Quill TE, Brody H. Physician recommendations and patient autonomy: finding a balance between physician power and patient choice. *Ann Intern Med.* 1996;125:763-769.
- 8. Grilli R, Ramsay C, Minozzi S. Mass media interventions: effects on health services utilisation. *Cochrane Database Syst Rev.* 2002;1:CD000389. Available at: http://www.mrw.interscience.wiley.com/cochrane/clsysrev/articles/CD000389/pdf\_fs.html. Accessed February 12, 2007.
- 9. Sharma V, Dowd MD, Swanson DS, Slaughter AJ, Simon SD. Influence of the news media on diagnostic testing in the emergency department. *Arch Pediatr Adolesc Med.* 2003;157:257-260.
- 10. Haas JS, Kaplan CP, Gerstenberger EP, Kerlikowske K. Changes in the use of postmenopausal hormone therapy after the publication of clinical trial results. *Ann Intern Med.* 2004;140:184-188.
- 11. Sulmasy DP. The impact of the media coverage on practice. *J Gen Intern Med*. 1996;11:62.
- 12. Moynihan R, Bero L, Ross-Degnan D, et al. Coverage by the news media of the benefits and risks of medications. *N Engl J Med.* 2000;342:1645-1650.
- 13. Pribble JM, Goldstein KM, Fowler EF, Greenberg MJ, Noel SK, Howell JD. Medical news for the public to use? What's on local TV news. *Am J Manag Care*. 2006;12:170-176.
- 14. Medical Professionalism Project. Medical professionalism in the new millennium: a physician charter. *Ann Intern Med.* 2002;136:243-245.

Vidya Sharma, MBBS, MPH, is a professor of pediatrics at the University of Missouri School of Medicine and is a staff pediatrician at the Children's Mercy Hospital and Clinics in Kansas City, Missouri.

M. Denise Dowd, MD, MPH, is a professor of pediatrics at the University of Missouri School of Medicine and the chief of the Section of Injury Prevention at the Children's Mercy Hospital and Clinics in Kansas City, Missouri.

#### Related article

Media reporting and emergency room testing trends, March 2007

The people and events in this case are fictional. Resemblance to real events or to names of people, living or dead, is entirely coincidental.

The viewpoints expressed on this site are those of the authors and do not necessarily reflect the views and policies of the AMA.

Copyright 2007 American Medical Association. All rights reserved.

169